Author’s response to reviews

Title: Pralatrexate as a Bridge to Allogeneic Hematopoietic Stem Cell Transplantation: a Case Report on a Patient with Advanced-stage Extranodal Nasal-type Natural killer/T-cell Lymphoma Refractory to First-line SMILE Chemotherapy

Authors:
Yao-Chung Liu (ycliu17@gmail.com)
Hao-Yuan Wang (b101091114@tmu.edu.tw)
Ting-An Lin (talin@vghtpe.gov.tw)
Po-Shen Ko (psko@vghtpe.gov.tw)
Chia-Jen Liu (chiajenliu@gmail.com)
Liang-Tsai Hsiao (lthsiao@vghtpe.gov.tw)
Sheng-Hsuan Chien (shchieu@vghtpe.gov.tw)
Jyh-Pyng Gau (jpgau@vghtpe.gov.tw)

Version: 2 Date: 08 Jan 2020

Author’s response to reviews:

Reply to reviewer comments:

Reviewer reports:

I have reviewed the revision. Although the authors responded well to the comments by the editor and the reviewers, they did not include these details within the text of the revision. They should also include these modifications within the text.

Answer: We will add the revised descriptions to the manuscript.